Author's response to reviews

Title: Insomnia among patients with advanced disease during admission in a Palliative Care Unit: a prospective observational study on its frequency and association with psychological, physical and environmental factors

Authors:

Anna Renom-Guiteras (Anna.Renom@uni-wh.de)
José Planas (17037@parcdesalutmar.cat)
Cristina Farriols (CFarriols@parcdesalutmar.cat)
Sergi Mojacal (seergi@gmail.com)
Ramón Miralles (RMiralles@parcdesalutmar.cat)
Maria A Silvent (94546@parcdesalutmar.cat)
Ada I Ruiz-Ripoll (IRuiz@parcdesalutmar.cat)

Version: 5 Date: 1 June 2014

Author's response to reviews: see over
Insomnia among patients with advanced disease during admission in a Palliative Care Unit: a prospective observational study on its frequency and association with psychological, physical and environmental factors.

Response to the Editorial Request

Thank you for this opportunity to improve the manuscript. Please find our answer below.

Editorial Request:
Only in Table 1, line Pfeiffer Index it should be 1.3 instead of 1.28. I think the authors have overseen this.

Authors’ answer:
Thank you for this minor revision, which we had overseen. We have done the suggested correction in Table 1.

Changes in the manuscript:
See Table 1.

With best wishes,
Anna Renom-Guiteras on behalf of all co-authors